Author's response to reviews

Title: A primary leptomeningeal histiocytic sarcoma with a good outcome: Case report and review of the literature

Authors:

Elisabeth Pérez Ruiz (elperu@hcs.es)
Mª Teresa Delgado (maitechu22@hotmail.com)
Andrés Sanz (andres.sanz.sspa@juntadeandalucia.es)
Ana Serradilla (anaserradilla@hotmail.com)
Antonio Rueda (arueda@hcs.es)

Version: 3 Date: 19 March 2013

Author's response to reviews:

Thank so much for your review. Our comment are the follow:

- we have included some IHC markers that we were included in the discussion but we have forgotten in the case presentation.
- we have tried to explain why we chose chemo radiotherapy adjuvant in this case. It is impossible to present convincing evidence and we thought that this treatment could be benefit. In cases with extra nodal disease, chemotherapy was benefit.
- We included the paper suggested by the reviewer. Radiation alone would be beneficial as the case reported by Wu et al. but our case occurred three years before. Moreover, we do not know whether radiation induced histiocytic sarcomas differ in their biological behavior.
- Sorry for the wrong spellings and grammatic errors!. We have corrected them